Title: Efficacy of Epidermal Growth Factor Receptor targeting in advanced chordoma: case report and literature review

Authors:

Simon Launay (launays@marseille.fnclcc.fr)
Bruno Chetaille (chetailleb@marseille.fnclcc.fr)
Fanny Medina (medinaf@marseille.fnclcc.fr)
Delphine Perrot (louveld@marseille.fnclcc.fr)
Serge Nazarian (Serge.nazarian@ap-hm.fr)
Jérôme Guiramand (guiramandj@marseille.fnclcc.fr)
Laurence Moureau-Zabotto (moureaul@marseille.fnclcc.fr)
François Bertucci (bertuccif@marseille.fnclcc.fr)

Version: 5 Date: 20 September 2011

Author's response to reviews: see over
Dear Editor,

Please, find enclosed the revised proofs of our paper by Launay et al entitled “Efficacy of Epidermal Growth Factor Receptor targeting in advanced chordoma: case report and literature review”, accepted for publication in BMC Cancer as a Case Report.

We thank your editorial board for the comments that have been taken into account.

1/ Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

Done

2/ Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

Done

3/ Abbreviations: Please format these in to a paragraph. Please make sure all abbreviations are spelt out and occur in alphabetical order.

Done

4/ Authors' contributions: please ensure this is formatted into a paragraph rather than a list or in bullet points

Done

5/ Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.

Done

6/ Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions.

Done

Thank you again for your help. Sincerely yours,

Pr François BERTUCCI